Author's response to reviews

Title: Transcriptional profiling of mycobacterial antigen-induced responses in infants vaccinated with BCG at birth

Authors:

Helen A Fletcher (helen.fletcher@ndm.ox.ac.uk)
Alana Keyser (Alana.Keyser@uct.ac.za)
Mark Bowmaker (M.Bowmaker@uct.ac.za)
Peter C Sayles (psayles@trudeauinstitute.org)
Gilla Kaplan (kaplangi@umdnj.edu)
Greg Hussey (ghussey@rmh.uct.ac.za)
Adrian VS Hill (adrian.hill@ndm.ox.ac.uk)
Willem A Hanekom (Willem.Hanekom@uct.ac.za)

Version: 4 Date: 11 February 2009

Author's response to reviews: see over
We have made the following formatting changes requested by the editorial office:

1. Table vertical elements - removed
2. Table size - These must fit into an A4 portrait page – changed to portrait
3. Figure title – Figure titles removed
4. Figure as two files – Figures merged to one file
5. Additional files – Additional files referred to correctly
6. Title page - Remove the text 'BMC MS 1962304890208351. Revised after reviewers' comments.' - removed
7. Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system - corrected
8. Please change 'Author Contributions' to 'Authors' contributions' – changed as requested
9. Figure cropping – figures have been cropped as far as possible